Author's response to reviews

Title: Healthcare-associated infections in Pediatric Cancer Patients: Results of a Prospective Surveillance Study from University Hospitals in Germany and Switzerland

Authors:

Arne Simon (asimon@ukb.uni-bonn.de)
Roland A Ammann (roland.ammann@insel.ch)
Udo Bode (udo.bode@ukb.uni-bonn.de)
Gudrun Fleischhack (gudrun.fleischhack@ukb.uni-bonn.de)
Hans-Martin Wenchel (hans-martin.wenchel@uk-koeln.de)
Dorothee Schwammborn (dorothee.schwammborn@uk-koeln.de)
Chara Gravou (Chara.Gravou@kinder.imed.uni-erlangen.de)
Paul Gerhardt Schlegel (schlegel_p@kinderklinik.uni-wuerzburg.de)
Stefan Rutkowski (rutkowski_s@kinderklinik.uni-wuerzburg.de)
Claudia Dannenberg (paediatrie@medizin.uni-halle.de)
Dieter Koerholz (paediatrie@medizin.uni-halle.de)
Hans Juergen Laws (laws@med.uni-duesseldorf.de)
Michael H Kramer (OXM3@aol.com)

Version: 5 Date: 18 May 2008

Author's response to reviews: see over
Dear colleagues,

Thank you for accepting our original article MS 982540511672993

**Healthcare-associated infections in Pediatric Cancer Patients: Results of a Prospective Surveillance Study from University Hospitals in Germany and Switzerland**


For publication in BMC Infectious Diseases.

**We performed the last formatting changes as requested in your mail from May 16th, 2008.**

Best regards on behalf of all authors

Arne Simon

Arne Simon MD
Children’s Hospital Medical Center, University of Bonn
Pediatric Hematology and Oncology
Adenauerallee 119
53113 Bonn
Tel 0049 22828733254
Fax 0049 22828733301

Email asimon@ukb.uni-bonn.de